



### **RESEARCH LETTER**

10.1029/2018GL078820

#### **Special Section:**

The Arctic: An AGU Joint Special Collection

#### **Key Points:**

- Arctic tundra vegetation "greening" (since 1982), as assessed by the Normalized Difference Vegetation Index (NDVI), increases from north to south
- The greatest interannual variability in tundra vegetation (NDVI) occurs in the midlatitudes of the tundra biome
- The annual relationship between summer warmth and tundra vegetation (NDVI) is strongest also in the midlatitudes of the tundra biome.

#### **Supporting Information:**

- Figure S1
- Figure S2
- Figure S3

#### Correspondence to:

L. M. Reichle, lmr8ws@virginia.edu

#### Citation:

Reichle, L. M., Epstein, H. E., Bhatt, U. S., Raynolds, M. K., & Walker, D. A. (2018). Spatial heterogeneity of the temporal dynamics of arctic tundra vegetation. *Geophysical Research Letters*, 45, 9206–9215. https://doi.org/10.1029/ 2018GL078820

Received 21 NOV 2017 Accepted 20 AUG 2018 Accepted article online 27 AUG 2018 Published online 12 SEP 2018

©2018. American Geophysical Union. All Rights Reserved.

# Spatial Heterogeneity of the Temporal Dynamics of Arctic Tundra Vegetation

L. M. Reichle<sup>1</sup>, H. E. Epstein<sup>1</sup>, U. S. Bhatt<sup>2</sup>, M. K. Raynolds<sup>3</sup>, and D. A. Walker<sup>4</sup>

<sup>1</sup>Department of Environmental Sciences, University of Virginia, Charlottesville, VA, USA, <sup>2</sup>Department of Atmospheric Sciences, University of Alaska Fairbanks, Fairbanks, AK, USA, <sup>3</sup>Alaska Geobotany Center, University of Alaska Fairbanks, Fairbanks, AK, USA, <sup>4</sup>Institute of Arctic Biology, University of Alaska Fairbanks, Fairbanks, AK, USA

**Abstract** Arctic tundra vegetation has largely been "greening" in recent decades, resulting in major changes to terrestrial ecosystems, with implications for surface energy balance, permafrost, carbon and water cycling, herbivore populations, and human land use. While general greening trends have been well-studied, more specific vegetation-temperature dynamics are spatially and temporally heterogeneous and currently not well understood. This study uses Normalized Difference Vegetation Index (NDVI) and Summer Warmth Index (SWI) data to investigate patterns of arctic tundra vegetation and temperature dynamics over North American and Eurasian continents and by Arctic bioclimate subzones (essentially latitudinal-based). Relative vegetation increases in northern subzones were muted compared to temperature than, relative temperature changes. Detrended, interannual NDVI variances were greatest in middle and southern subzones, whereas interannual SWI variances were greatest in southern subzones. Annual SWI and NDVI relationships were strongest in midlatitude subzones.

**Plain Language Summary** Arctic tundra vegetation has generally been increasing in biomass and productivity in recent decades. This paper uses 34 years of satellite data to investigate more thoroughly the relationships between temperature and vegetation in the tundra over time and space. We found that in the southern part of the tundra biome, the vegetation (as indicated by the satellite data) increased at the same rate as, or faster than, temperature, whereas northern tundra vegetation increased more slowly than temperature. The variation in tundra vegetation from year to year was greatest in the middle and southern parts of the Arctic tundra biome, whereas temperature variation from year to year increased from north to south. On an annual basis, the strongest relationships between temperature and vegetation occurred in the midlatitudes of the tundra biome.

#### 1. Introduction

The Arctic has experienced increasing temperatures at an accelerated rate in recent decades, and this trend is expected to continue throughout this century (Kirtman et al., 2014; Overland et al., 2016; Stocker, 2014). Warming has been observed up to 2–3 times greater in the Arctic compared to average global temperature increases (Kaufman et al., 2009). Consequences of warming in the Arctic include loss of sea ice and changes to permafrost, land surface hydrology, vegetation, and biodiversity, as well as alterations of human land use (e.g., ACIA, 2004; Richter-Menge et al., 2017; Richter-Menge & Mathis, 2017). The Arctic has already experienced major vegetation changes, including a general "greening" (increased vegetation productivity) of the arctic tundra (e.g., Bhatt et al., 2010, 2013; Goetz et al., 2005; Jia et al., 2003). Dynamics in vegetation have commonly been observed using the Normalized Difference Vegetation Index (NDVI), a satellite-derived index of green vegetation (Tucker & Sellers, 1986). Satellite-derived NDVI is determined based on the solar radiation absorbed by the process of photosynthesis (predominantly red wavelengths) and that reflected by leaf structural and chemical components (near infrared) and provides a reliable index for the quantity of green vegetation.  $NDVI = \frac{NIR-Red}{NIR+Red}$ , where NIR is reflectance in near infrared radiation (725–1,000 nm), and Red is reflectance in visible red radiation (580–680 nm).

Recent studies have evaluated dynamics of coastally influenced, circumpolar arctic tundra vegetation. Bhatt et al. (2013) examined NDVI, Summer Warmth Index (SWI, sum of mean monthly temperatures >0 °C), and coastal sea ice (among other variables) of the Arctic from 1982–2011. The Eurasian Arctic experienced a 10% increase in MaxNDVI (peak annual value) from 1982 to 2011, whereas time-integrated NDVI (TI-NDVI;

time-integrated summer values) increased 10% from 1982 to 2005, before it began declining. The eastern North American Arctic experienced slight increases in MaxNDVI from 1982 to 2009 and greater increases from 2009 to 2011 for an overall 15% increase; TI-NDVI in eastern North America increased overall by 8%. The western North American Arctic experienced a steady increase in TI-NDVI (overall 8% increase) and an accelerated increase in MaxNDVI since 2005 (18% increase from 1986 to 2011; Bhatt et al., 2013).

Despite these and other observation of changes in tundra (Bhatt et al., 2017; Ju & Masek, 2016; Xu et al., 2013), the environmental controls on tundra vegetation dynamics (e.g., NDVI) are still not very well understood, although certain potential driving variables have been explored. Bhatt et al. (2010, 2013) found negative correlations between NDVI and Arctic sea ice extent (i.e., less sea ice was related to greater NDVI). Macias-Fauria et al. (2012) found that vegetation growth during the peak growing season responded strongly to the position of continental air mass in the Eurasian Arctic. Bieniek et al. (2015) investigated biweekly NDVI values of the Alaskan tundra in three locations (coastal areas of the Beaufort, East Chukchi, and East Bering Seas) from 1982 to 2013. In all three regions, lower NDVI values during the early growing season were related to increasing spring snow depth and potentially later snowmelt (Bieniek et al., 2015). Another study investigating long-term NDVI trends in Alaskan tundra found a strong correlation between NDVI and the SWI from the previous year (Verbyla, 2008).

Previous studies on tundra vegetation dynamics have illustrated the substantial spatial (and temporal) heterogeneity of arctic tundra greening trends. Earlier studies of tundra greening suggested that this phenomenon was largely due to tall shrub expansion in riparian areas (Sturm et al., 2001). However, Frost and Epstein (2014) found greening in Siberia to be occurring in upland areas as well, also as a result of tall shrub (alder) expansion. The greening (NDVI increases) observed was occurring more rapidly in areas of expanding shrublands than those of existing shrub tundra (Frost et al., 2014), facilitated by disturbances to other vegetation, such as fires, landslides, and frost heave (Frost et al., 2013). A study of the Canadian Arctic over recent decades related spatiotemporal NDVI trends to plant functional type composition. The greatest greening occurred in tundra subzones C and D (midsouthern tundra; based on the Circumpolar Arctic Vegetation Map, CAVM Team 1995), where dominant sedges may have responded rapidly to interannual warming (Jia et al., 2009).

While arctic tundra greening patterns (e.g., NDVI trends) as a whole have been widely investigated, few studies have rigorously evaluated in detail the spatial and temporal heterogeneity of temperature-vegetation relationships throughout the circumpolar Arctic. An understanding of how these patterns of greening vary systematically with latitude and by continent is a next step in determining the spatial variability in tundra vegetation dynamics and perhaps elucidating different environmental controls on tundra vegetation productivity. This study investigates long-term trends in SWI and NDVI over the length of the remote sensing record (1982–2015) and how they vary by tundra bioclimate subzone and continent (North America and Eurasia). It also explores interannual variabilities of SWI and NDVI by tundra bioclimate subzone and continent. In addition, possible controls of arctic tundra NDVI are evaluated through investigating annual relationships between SWI and NDVI.

#### 2. Materials and Methods

Satellite-derived, biweekly composited NDVI data were acquired from the NASA GIMMS3g data set developed largely from Advance Very High Resolution Radiometer sensors flown on National Oceanic and Atmospheric Administration satellites (Pinzon & Tucker, 2014). Two NDVI-related variables were calculated as indices for vegetation activity: MaxNDVI and TI-NDVI. MaxNDVI is the maximum annual NDVI value (essentially the summer peak), thereby depicting peak photosynthetic capacity. TI-NDVI is the sum of biweekly NDVI values greater than 0.05 from May to September, thereby depicting the total vegetation activity and indicative of net primary production throughout the growing season. All analyses described herein referring to NDVI were done for both MaxNDVI and TI-NDVI. SWI data were determined from remotely sensed Land Surface Temperature data (also from the Advance Very High Resolution Radiometer record; Comiso, 2003) as the sum of mean monthly surface temperatures greater than 0 °C (°C months). Data were separated by continent and Arctic bioclimate subzone (A-E) from 1982 to 2015. The bioclimate subzones of the Circumpolar Arctic Vegetation Map categorize the tundra essentially latitudinally, with Subzone A being the coldest, highest latitude and Subzone E being the warmest, lowest latitude nonalpine tundra (Walker et al., 2005).

To compare the temporal trends between temperature and vegetation change, SWI and NDVI values for each pixel were averaged for each year by bioclimate subzone and continent, and then trends were calculated.



**Figure 1.** Summary of MaxNDVI and Summer Warmth Index (SWI) *z* scores by bioclimate subzone (A through E) for North America and Eurasia (only significant regression lines plotted); summary graph of regression slopes (the positive values indicate that MaxNDVI is increasing relatively faster than SWI; the negative values indicate that SWI is increasing relatively faster than MaxNDVI); tables of *P* values and slopes of linear regressions by bioclimate subzone and continent.

Linear regression analyses compared normalized SWI and NDVI values over time for the various spatial classifications of tundra (subzone and continent). Data were normalized using *z* scores, allowing for more direct comparison of temporal trends in NDVI and SWI, which vary in units, magnitudes, and starting values. This approach also minimized the effect of possibly anomalous starting values. The difference between slopes (*m* values) of these trend lines (NDVIm – SWIm) indicated whether vegetation (NDVI) or temperature (SWI) was increasing (or decreasing) by larger relative magnitudes. Positive values of slope differences indicated that NDVI increased relatively faster than SWI over the 34-year period, whereas negative values indicated that SWI increased relatively faster than NDVI (Figure 1).

To examine interannual variability of SWI and NDVI by subzone and continent irrespective of the trends, first, the raw data were detrended. Residuals were calculated as the difference between the annual linear trend values and the actual data points for both SWI and NDVI. Variances of residuals were used to compare interannual variabilities of SWI and NDVI across bioclimate subzones and continents, irrespective of any directional trend. Coefficients of variation were also calculated to compare relative variabilities, accounting for differences in mean values.

Finally, several analyses were conducted to examine possible controls of NDVI in any given year or what controls NDVI change from year to year. First, a general linear regression of NDVI as a function of SWI was conducted to determine direct controls of cumulative summer warmth on NDVI by bioclimate subzone and continent. Second, to examine the extent to which the temperature of a given year correlated with vegetation change from the prior year, a linear regression analysis between SWI and the difference between the NDVI of that year and the NDVI of the prior year was conducted for each bioclimate subzone and continent. A third analysis compared annual changes in SWI and NDVI. Differences between the SWI of 1 year and the SWI of the prior year were regressed against the differences between the NDVI of 1 year and the NDVI of the prior year. Slopes of these regressions were used to evaluate the responses of NDVI to SWI by bioclimate subzones and continents. A final linear regression analysis evaluated the detrended NDVI of 1 year as a function of the NDVI of the previous year for each bioclimate subzone and continent, demonstrating to what degree the vegetation of 1 year was correlated with the vegetation of the prior year, irrespective of any trend. *P* values are calculated for all regressions, and only trend lines of significant regressions (*P* values < 0.05) were displayed in figures.

#### 3. Results

Regressions of the detrended MaxNDVI/TI-NDVI of 1 year and the MaxNDVI/TI-NDVI of the next year showed no significant relationships for any subzone/continent combination. Therefore, irrespective of any trend, there is no relationship between NDVI in a given year and NDVI of the following year, and therefore, the individual years may be considered as independent samples.

The differences between normalized MaxNDVI changes and SWI changes were lowest (most negative) in subzone A for Eurasia and in subzone B for North America (Figure 1). In subzone A of both North America and Eurasia, there were significant positive trends in SWI, but no significant trend in MaxNDVI. Differences between the normalized changes in MaxNDVI and SWI increased (and switched from negative to positive) with decreasing latitude (with the exception of subzone B in North America). Thus, at high latitudes (subzones A and B), the SWI normalized changes were greater than those for MaxNDVI for both continents. At lower latitudes (subzones D and E), the MaxNDVI normalized changes were greater than those for SWI for both continents (differences between MaxNDVI and SWI regression slopes range from 0.01 to 0.09). In contrast to what was observed for subzone A, both North America and Eurasia demonstrate significant positive trends in MaxNDVI for subzone E, but no significant trends in SWI. The Eurasian continent had more positive differences between the slopes of MaxNDVI and SWI relative change across all subzones and fewer significant trends in SWI compared to North America (Figure 1).

Most TI-NDVI trends were consistent with MaxNDVI relative changes. The most notable disparity was the overall more negative difference between regression slopes of normalized TI-NDVI and SWI compared to MaxNDVI (i.e., SWI increasing to a greater extent relative to TI-NDVI; Figure S1 in the supporting information). Further, North America demonstrated more positive differences between regression slopes of TI-NDVI and SWI compared to Eurasia (with the exception of subzone B), which contrasts the pattern seen in MaxNDVI. Subzone B of North America experienced the most negative difference between regression slopes of TI-NDVI and SWI relative change. Additionally, there were fewer significant trends in TI-NDVI compared to MaxNDVI, especially in Eurasia (significantly negative trend only for subzone A).

Summer Warmth Index generally had greater detrended variances in southern subzones, which were greater for North America than Eurasia (Figure 2a). However, because of the greater magnitude of SWI variances, there were no statistically different values across bioclimate subzones and continents (P > 0.05). MaxNDVI variances were greatest in midsubzones, with both Eurasia and North America experiencing the greatest variances in subzone C. Eurasia had much higher variation in subzones B and C, slightly lower variation in subzone D, and very similar variation in subzones A and E compared to North America. For TI-NDVI, Eurasia variances were much greater than those for North America, except for subzone A, where they were similar. In both continents, the variance increased through subzones A, B, and C. In Eurasia, variances were slightly lower in subzones D and E compared to the rest of the continent. In North America, variance remained relatively constant through subzones C, D, and E.

The coefficients of variation of SWI, MaxNDVI, and TI-NDVI generally decreased from north to south. North America had greater coefficients of variation than Eurasia in both subzones A and B for SWI, MaxNDVI, and TI-NDVI, but not all of these differences were significant (P > 0.05). MaxNDVI and TI-NDVI had very similar values for coefficients of variation across all subzones, especially for North America (Figure 3).



a)

SWI variance



**Figure 2.** Summary of residual variances of (a) Summer Warmth Index (SWI), (b) MaxNDVI, and (c) time-integrated Normalized Difference Vegetation Index from 1982 to 2015 by bioclimate subzone (A through E); the roman numerals indicate variances that are not significantly different (P > 0.05) on (b) and (c); there were no significant differences between SWI variances.

The regressions of SWI and MaxNDVI by subzone and continent were strongest with significant, positive relationships in the mid subzones, whereas more northern and southern subzones showed weaker, less significant relationships. There is essentially no relationship between SWI and MaxNDVI in subzones A, B, and E of North America. However, there is a positive, significant correlation in subzone C ( $r^2 = 0.36$ ) and a weaker, but still significant, positive correlation in subzone D ( $r^2 = 0.15$ ) over the 34-year period. There is also no significant relationship between SWI and MaxNDVI in subzones D and E of Eurasia. The strongest and most positive correlation between SWI and MaxNDVI in Eurasia occurs in subzone B ( $r^2 = 0.23$ ). Subzones A and C of Eurasia show weaker, but significant, positive correlations ( $r^2 = 0.18$  and 0.16, respectively; Figure 4).

Regressions of SWI and TI-NDVI exhibited strong, positive relationships in midsouthern subzones. There were no significant correlations in subzones A, B, and E on either continent. In Eurasia, subzone D had a weak, but significant, positive correlation ( $r^2 = 0.13$ ). Subzone C had a slightly stronger significant positive correlation





 $(r^2 = 0.17)$ . In North America, subzones C and D had relatively strong significant positive correlations ( $r^2 = 0.23$  and 0.21, respectively; Figure S2).

These general trends were consistent with regressions of SWI and the difference between MaxNDVI/TI-NDVI of 1 year and MaxNDVI/TI-NDVI of the prior year. The strongest, most positive relationships in Eurasia occurred in subzone B, and these declined in the more southern subzones. The strongest, most positive relationships in North America occurred in subzone C, and these declined in more northern and southern subzones. Eurasia also had overall more positive relationships. These trends also were consistent with regressions of the difference between SWI of 1 year and the prior year versus the difference between MaxNDVI/TI-NDVI of one year and the prior year (results not shown).

#### 4. Discussion

The temporal dynamics of vegetation across Arctic latitudes and continents could provide essential information for understanding the processes driving current tundra greening/browning patterns, as well as being able



Figure 4. Regression of Summer Warmth Index and MaxNDVI for each bioclimate subzone (A through E) for North America and Eurasia (only significant trends plotted); summary graph of the regression slopes; table of *P* values of linear regressions by bioclimate and subzone.

to predict future trends. Vegetation changes affect land-surface albedo, permafrost thaw, animal migration patterns, carbon sequestration, human activities, and more (e.g., Fauchald et al., 2017; Frost et al., 2017; Horstkotte et al., 2017; Kępski et al., 2017). This study has identified (1) the dynamic patterns of arctic tundra vegetation and (2) the degree to which summer warmth drives tundra vegetation change, and results will provide inputs to analyses that evaluate vegetation change effects on other tundra properties.

From a spatial perspective, throughout the arctic tundra, there is a strong positive relationship between summer temperatures and both NDVI and aboveground tundra biomass (Epstein et al., 2008; Raynolds et al., 2012; Walker et al., 2012), and therefore, a positive temporal correlation between SWI and NDVI may be expected. However, the temporal relationship between summer temperatures and vegetation productivity in arctic tundra is likely complex, with mixed findings in the literature. Numerous field studies suggest that interannual variability in tundra productivity is generally constrained and not strongly linked to summer temperatures (e.g., Chapin III & Shaver, 1985; Hudson & Henry, 2010), although others show positive relationships between summer temperatures and tundra plant growth (Callaghan et al., 1989; Van der Wal & Stien, 2014). In our coarse-scale remote sensing analysis, we did not find consistent relationships between NDVI and SWI (integrated summer warmth), and there are potentially several reasons why increasing SWI over time might not lead to concomitant increases in NDVI. One reason is that increasing summer warmth could lead to melting of ground ice and/or thawing of permafrost, the development of thermokarst, and the ponding of surface water, which would decrease the landscape NDVI signal (at least at the resolutions of many current satellite sensors; Liljedahl et al., 2016; Raynolds et al., 2014). Increases in fall and winter temperatures (likely correlated with SWI) can potentially lead to increases in snow depth, a later melt of the deeper snowpack, and a reduced

100

spring NDVI (Bhatt et al., 2017). Additionally, the melting of coastal sea ice with increases in SWI will reduce the degree of continentality of the adjacent landmasses, with complex indirect effects on regional temperatures and the productivity of vegetation (Macias-Fauria et al., 2017).

The results of this study indicate that temporal relationships between SWI and NDVI are strongest in middle and more southern subzones of the Arctic. In northern subzones, SWI increases to a relatively greater degree than both MaxNDVI and TI-NDVI over the 34-year period (Figures 1 and 2). The muted increase in NDVI compared to SWI seen in northern latitudes is consistent with other Arctic remote sensing (Vickers et al., 2016) and field (Bjorkman et al., 2017; Edwards et al., 2016) studies. There is less vascular vegetation biomass and greater relative nonvascular vegetation biomass as latitude increases (Epstein et al., 2008), and these types are likely to exhibit different responses to changing temperatures. In addition, delays in vegetation response are due to growth constraints of the existing small, cold-adapted plants, as well as limits on reproduction and colonization by new species from warmer areas. In more southern subzones, experiencing equivalent or greater increases in NDVI than SWI, vegetation increase is potentially the effect of positive feedback mechanisms (e.g., more vegetation leads to greater growth). Additionally, increased vegetation decreases albedo (particularly in the spring), which leads to an increase in sensible heat flux and warmer near-surface air temperatures (Chapin III et al., 2005; Lawrence & Swenson, 2011), promoting vegetation growth. Greater vegetation increases relative to SWI increases can also be the result of climate-related disturbances, such as fire, and landslides, which expose substrate for shrub expansion (e.g., Jin et al., 2012; Lantz et al., 2013; Liebman et al., 2015).

Elmendorf et al. (2012) found that tall shrubs (characteristic of southern subzones) increased in warming experiments, while dwarf shrubs (more characteristic of midlatitude tundra subzones) decreased under experimental warming. Thus, it was predicted that most shrub expansion would occur in subzones D and E, while higher latitude regions would be resistant to shrub expansion. Tall shrub expansion has been observed to occur throughout the southern tundra subzones (e.g., Frost & Epstein, 2014; Myers-Smith et al., 2011; Sturm et al., 2001). These results, along with other studies finding increasing NDVI and "greening" with temperature in southern subzones (Epstein et al., 2012; Forbes et al., 2010; Pouliot et al., 2009), are consistent with our results. Interestingly, Lloyd et al. (2011) found increased productivity with warming only in northern Siberian taiga rather than more southern tundra, northern taiga, and their ecotone.

Regressions of SWI and NDVI indicate that midsubzone (B and C) vegetation is more highly correlated with SWI than higher and lower latitudes on an annual scale. Further, our data demonstrate that NDVI generally correlates to a greater extent with SWI in Eurasia compared to North America. Similar conclusions were made by Myers-Smith et al. (2015), who found climate sensitivity of shrub growth heterogeneous, with greater sensitivity in northern Eurasian sites and sites with higher soil moisture. Myers-Smith et al. (2015) further found that, by latitude, climate sensitivity of growth was greatest at the boundary between the Low and High Arctic. Prevéy et al. (2017), on the other hand, found that flowering and leaf emergence were more sensitive to temperature in colder, more northern latitudes than warmer Arctic sites on an annual basis. However, temperature sensitivity in the date of a phenological event may not be related to the sensitivity of an index of vegetation quantity (e.g., NDVI). Finally, the lack of any correlation between detrended NDVI and the prior year values shows that vegetation quantity in a given year does not affect vegetation in the following year beyond the current directional trend; in other words, there is minimal, if any, temporal autocorrelation (1-year lag) in the detrended NDVI. Examining the 1-year lag autocorrelation in the NDVI data, including any trend, shows no evidence of any accelerating trends; in fact, many of the subzone-continent combinations indicated potential negative feedbacks (results not shown).

In summary, vegetation (NDVI) and temperature (SWI) trends from 1982 to 2015 differed substantially with latitude (and somewhat with continent); relative temperature increases were much greater than relative vegetation increases in the northernmost tundra regions (subzone A), and this difference decreased with decreasing latitude (i.e., further south), even to the extent where NDVI had a greater relative increase than SWI. Relative interannual variabilities in NDVI and the interannual responses of NDVI to SWI were greatest in midtundra subzones (e.g., subzones B–D), compared to the furthest north and furthest south tundra regions. These results are extremely valuable for understanding how a heterogeneous arctic tundra biome is responding to climate dynamics, and for how these systems might continue to change in the future.



#### Acknowledgments

This study was supported by NASA/Northern Eurasia Earth Science Partnership Initiative (NEESPI) Land Cover Land Use Change (LCLUC) Program, synthesis grant NNX14AD90. The data are already available at https:// ecocast.arc.nasa.gov/data/pub/gimms/.

#### References

ACIA (2004). Arctic climate impact assessment. Cambridge: Cambridge University Press.

- Bhatt, U. S., Walker, D. A., Raynolds, M. K., Bieniek, P. A., Epstein, H. E., Comiso, J. C., et al. (2013). Recent declines in warming and vegetation greening trends over pan-arctic tundra. *Remote Sensing*, *5*(9), 4229–4254. https://doi.org/10.3390/rs5094229
- Bhatt, U. S., Walker, D. A., Raynolds, M. K., Bieniek, P. A., Epstein, H. E., Comiso, J. C., et al. (2017). Changing seasonality of panarctic tundra vegetation in relationship to climatic variables. *Environmental Research Letters*, 12(5), 055003. https://doi.org/10.1088/1748-9326/aa6b0b Bhatt, U. S., Walker, D. A., Ravnolds, M. K., Comiso, J. C., Epstein, H. E., Jia, G., et al. (2010). Circumpolar arctic tundra vegetation change is linked
- to sea ice decline. *Earth Interactions*, 14(8), 1–20. https://doi.org/10.1175/2010El315.1 Bieniek, P. A., Bhatt, U. S., Walker, D. A., Raynolds, M. K., Comiso, J. C., Epstein, H. E., et al. (2015). Climate drivers linked to changing seasonality
- of Alaska coastal tundra vegetation productivity. *Earth Interactions*, *19*(19), 1–29. https://doi.org/10.1175/EI-D-15-0013.1 Bjorkman, A. D., Vellend, M., Frei, E. R., & Henry, G. H. (2017). Climate adaptation is not enough: Warming does not facilitate success of
- southern tundra plant populations in the high arctic. *Global Change Biology*, 23(4), 1540–1551. https://doi.org/10.1111/gcb.13417 Callaghan, T. V., Carlsson, B. A., & Tyler, N. J. C. (1989). Historical records of climate-related growth in Cassiope tetragona from the Arctic.
  - Journal of Ecology, 77(3), 823–837. https://doi.org/10.2307/2260988
- Chapin, F. S. III, & Shaver, G. S. (1985). Individualistic growth response of tundra plant species to environmental manipulations in the field. *Ecology*, 66(2), 564–576. https://doi.org/10.2307/1940405
- Chapin, F. S. III, Sturm, M., Serreze, M. C., McFadden, J. P., Key, J. R., Lloyd, A. H., et al. (2005). Role of land-surface changes in Arctic summer warming. Science, 310(5748), 657–660. https://doi.org/10.1126/science.1117368
- Comiso, J. C. (2003). Warming trends in the Arctic from clear sky satellite observations. Journal of Climate, 16(21), 3498–3510. https://doi.org/ 10.1175/1520-0442(2003)016<3498:WTITAF>2.0.CO;2
- Edwards, M., Grosse, G., Jones, B. M., & McDowell, P. (2016). The evolution of a thermokarst-lake landscape: Late quaternary permafrost degradation and stabilization in interior Alaska. Sedimentary Geology, 340, 3–14. https://doi.org/10.1016/j.sedgeo.2016.01.018
- Elmendorf, S. C., Henry, G. H. R., Hollister, R. D., Bjork, R. G., Bjorkman, A. D., Callaghan, T. V., et al. (2012). Global assessment of experimental climate warming on tundra vegetation: Heterogeneity over space and time. *Ecology Letters*, 15(2), 164–175. https://doi.org/10.1111/ j.1461-0248.2011.01716.x
- Epstein, H. E., Raynolds, M. K., Walker, D. A., Bhatt, U. S., Tucker, C. J., & Pinzon, J. E. (2012). Dynamics of aboveground phytomass of the circumpolar Arctic tundra during the past three decades. *Environmental Research Letters*, 7(1), 015506. https://doi.org/10.1088/1748-9326/7/1/015506
- Epstein, H. E., Walker, D. A., Raynolds, M. K., Jia, G. J., & Kelley, A. M. (2008). Phytomass patterns across a temperature gradient of the North American arctic tundra. *Journal of Geophysical Research*, 113, G03S02. https://doi.org/10.1029/2007JG000555
- Fauchald, P., Park, T., Tømmervik, H., Myneni, R., & Helene Hauser, V. (2017). Arctic greening from warming promotes declines in caribou populations. *Scientific Advances*, *3*, e1601365. https://doi.org/10.1126/sciadv.1601365
- Forbes, B. C., Macias Fauria, M., & Zetterberg, P. (2010). Russian arctic warming and 'greening' are closely tracked by tundra shrub willows. Global Change Biology, 16(5), 1542–1554. https://doi.org/10.1111/j.1365-2486.2009.02047.x
- Frost, G. V., & Epstein, H. E. (2014). Tall shrub and tree expansion in Siberian tundra ecotones since the 1960s. *Global Change Biology*, 20(4), 1264–1277. https://doi.org/10.1111/gcb.12406
- Frost, G. V., Epstein, H. E., & Walker, D. A. (2014). Regional and landscape-scale variability of Landsat-observed vegetation dynamics in northwest Siberian tundra. *Environmental Research Letters*, 9(2), 025004. https://doi.org/10.1088/1748-9326/9/2/025004
- Frost, G. V., Epstein, H. E., Walker, D. A., Matyshak, G., & Ermokhina, K. (2013). Patterned-ground facilitates shrub expansion in low arctic tundra. *Environmental Research Letters*, 8(1), 015035. https://doi.org/10.1088/1748-9326/8/1/015035
- Frost, G. V., Epstein, H. E., Walker, D. A., Matyshak, G., & Ermokhina, K. (2017). Seasonal and long-term changes in active-layer temperatures after tall shrubland expansion and succession in Arctic tundra. *Ecosystems*, 16, 1296.
- Goetz, S. J., Bunn, A. G., Fiske, G. J., & Houghton, R. A. (2005). Satellite-observed photosynthetic trends across boreal North America associated with climate and fire disturbance. *Proceedings of the National Academy of Sciences of the United States of America*, 102(38), 13,521–13,525. http://doi.org/10.1073/pnas.0506179102
- Horstkotte, T., Utsi, T., Larsson-Blind, Å., Burgess, P., Johansen, B., Käyhkö, J., et al. (2017). Human-animal agency in reindeer management: Herders' perspectives on vegetation dynamics under climate change. *Ecosphere*, 8(9), e01931. https://doi.org/10.1002/ecs2.1931
- Hudson, J. M. G., & Henry, G. H. R. (2010). High Arctic plant community resists 15 years of experimental warming. *Journal of Ecology*, 98(5), 1035–1041. https://doi.org/10.1111/j.1365-2745.2010.01690.x
- Jia, G., Epstein, H., & Walker, D. (2003). Greening of arctic Alaska, 1981–2001. Geophysical Research Letters, 30(20), 2067. https://doi.org/ 10.1029/2003GL018268
- Jia, G. J., Epstein, H. E., & Walker, D. A. (2009). Vegetation greening in the Canadian arctic related to decadal warming. *Journal of Environmental Monitoring*, 11(12), 2231–2238. https://doi.org/10.1039/b911677j
- Jin, Y. F., Randerson, J. T., Goetz, S. J., Beck, P. S. A., Loranty, M. M., & Goulden, M. L. (2012). The influence of burn severity on postfire vegetation recovery and albedo change during early succession in North America boreal forests. *Journal of Geophysical Research*, 117, G01036. https://doi.org/10.1029/2011JG001886
- Ju, J., & Masek, J. G. (2016). The vegetation greenness trend in Canada and US Alaska from 1984–2012 Landsat data. *Remote Sensing of Environment*, 176, 1–16. https://doi.org/10.1016/j.rse.2016.01.001
- Kaufman, D. S., Schneider, D. P., McKay, N. P., Ammann, C. M., Bradley, R. S., Briffa, K. R., et al. (2009). Recent warming reverses long-term arctic cooling. Science, 325(5945), 1236–1239. https://doi.org/10.1126/science.1173983
- Kępski, D., Luks, B., Migala, K., Wawrzyniak, T., Westermann, S., & Wojtuń, B. (2017). Terrestrial remote sensing of snowmelt in a diverse High-Arctic tundra environment using time-lapse imagery. *Remote Sensing*, 9(7), 733. https://doi.org/10.3390/rs9070733
- Kirtman, B., Power, S. B., Adedoyin, A. J., Boer, G. J., Bojariu, R., Camilloni, I., et al. (2014). In T. F. Stocker, D. Qin, G. K. Plattner, M. M. B. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), Near-term climate change: Projections and predictability. Cambridge, England: Cambridge University Press.
- Lantz, T. C., Marsh, P., & Kokelj, S. V. (2013). Recent shrub proliferation in the Mackenzie Delta uplands and microclimatic implications. *Ecosystems*, 16(1), 47–59. https://doi.org/10.1007/s10021-012-9595-2
- Lawrence, D. M., & Swenson, S. C. (2011). Permafrost response to increasing arctic shrub abundance depends on the relative influence of shrubs on local soil cooling versus large-scale climate warming. *Environmental Research Letters*, 6(4), 045504. https://doi.org/10.1088/ 1748-9326/6/4/045504
- Liebman, M., Khomutov, A., Gubarkov, A., Mullanurov, D., & Dvornikov, Y. (2015). The research station "Vaskiny Dachi", central Yamal, West Siberia, Russia—A review of 25 years of permafrost studies. *Fennia International Journal of Geography*, 193, 3–30.

- Liljedahl, A. K., Boike, J., Daanen, R. P., Fedorov, A. N., Frost, G. V., Grosse, G., et al. (2016). Pan-Arctic ice-wedge degradation in warming permafrost and its influence on tundra hydrology. *Nature Geoscience*, 9(4), 312–318. https://doi.org/10.1038/ngeo2674
- Lloyd, A. H., Bunn, A. G., & Berner, L. (2011). A latitudinal gradient in tree growth response to climate warming in the Siberian taiga. Global Change Biology, 17(5), 1935–1945. https://doi.org/10.1111/j.1365-2486.2010.02360.x
- Macias-Fauria, M., Forbes, B. C., Zetterberg, P., & Kumpula, T. (2012). Eurasian arctic greening reveals teleconnections and the potential for structurally novel ecosystems. *Nature Climate Change*, 2(8), 613–618. https://doi.org/10.1038/NCLIMATE1558
- Macias-Fauria, M., Karlsen, S. R., & Forbes, B. C. (2017). Disentangling the coupling between sea ice and tundra productivity in Svalbard. *Scientific Reports*, 7(1), 8586. https://doi.org/10.1038/s41598-017-06218-8
- Myers-Smith, I. H., Elmendorf, S. C., Beck, P. S. A., Wilmking, M., Hallinger, M., Blok, D., et al. (2015). Climate sensitivity of shrub growth across the tundra biome. *Nature Climate Change*, 5(9), 887–891. https://doi.org/10.1038/NCLIMATE2697
- Myers-Smith, I. H., Forbes, B. C., Wilmking, M., Hallinger, M., Lantz, T., Blok, D., et al. (2011). Shrub expansion in tundra ecosystems: Dynamics, impacts, and research priorities. *Environmental Research Letters*, 6(4), 045509. https://doi.org/10.1088/1748-9326/6/4/045509
- Overland, J., Hanna, E., Hanssen-Bauer, I., Kim, S. J., Walsh, J. E., Wang, M., et al. (2016). Surface air temperature [Arctic essays]. Arctic Report Card 2016.
- Pinzon, J. E., & Tucker, C. J. (2014). A non-stationary 1981-2012 AVHRR NDVI3g time series. Remote Sensing, 6(8), 6929–6960. https://doi.org/ 10.3390/rs6086929
- Pouliot, D., Latifovic, R., & Olthof, I. (2009). Trends in vegetation NDVI from 1 km AVHRR data over Canada for the period 1985–2006. International Journal of Remote Sensing, 30(1), 149–168. https://doi.org/10.1080/01431160802302090
- Prevéy, J., Vellend, M., Rüger, N., Hollister, R. D., Bjorkman, A. D., Myers-Smith, I. H., et al. (2017). Greater temperature sensitivity of plant phenology at colder sites: Implications for convergence across northern latitudes. *Global Change Biology*, 23(7), 2660–2671. https://doi. org/10.1111/gcb.13619
- Raynolds, M. K., Walker, D. A., Ambrosius, K. J., Brown, J., Everett, K. R., Kanevskiy, M., et al. (2014). Cumulative geoecological effects of 62 years of infrastructure and climate change in ice-rich permafrost landscapes, Prudhoe Bay oilfield, Alaska. *Global Change Biology*, 20(4), 1211–1224. https://doi.org/10.1111/gcb.12500
- Raynolds, M. K., Walker, D. A., Epstein, H. E., Pinzon, J. E., & Tucker, C. J. (2012). A new estimate of tundra-biome phytomass from trans-arctic field data and AVHRR NDVI. *Remote Sensing Letters*, 3(5), 403–411. https://doi.org/10.1080/01431161.2011.609188

Richter-Menge, J., & Mathis, J. (2017). The Arctic: In state of the climate 2016. Bulletin of the American Meteorological Society, 98(8), S132–S134. Richter-Menge, J., Overland, J. E., Mathis, J. T., & Osborne, E. (2017). Arctic report card, 2017.

- Stocker, T. (Ed.) (2014). Climate change 2013: The physical science basis: Working group I contribution to the fifth assessment report of the Intergovernmental Panel on Climate Change. Cambridge: Cambridge University Press.
- Sturm, M., Racine, C., & Tape, K. (2001). Climate change—Increasing shrub abundance in the arctic. Nature, 411(6837), 546–547. https://doi. org/10.1038/35079180
- Tucker, C., & Sellers, P. (1986). Satellite remote-sensing of primary production. International Journal of Remote Sensing, 7(11), 1395–1416. https://doi.org/10.1080/01431168608948944
- Van der Wal, R., & Stien, A. (2014). High-arctic plants like it hot: A long-term investigation of between-year variability in plant biomass. *Ecology*, 95(12), 3414–3427.
- Verbyla, D. (2008). The greening and browning of Alaska based on 1982–2003 satellite data. *Global Ecology and Biogeography*, 17(4), 547–555 . https://doi.org/10.1111/j.1466-8238.2008.00396.x
- Vickers, H., Hogda, K. A., Solbo, S., Karlsen, S. R., Tommervik, H., Aanes, R., & Hansen, B. B. (2016). Changes in greening in the high arctic: Insights from a 30 year AVHRR max NDVI dataset for Svalbard. *Environmental Research Letters*, 11(10), 105004. https://doi.org/10.1088/ 1748-9326/11/10/105004
- Walker, D., Raynolds, M., Daniels, F., Einarsson, E., Elvebakk, A., Gould, W., et al. (2005). The circumpolar arctic vegetation map. *Journal of Vegetation Science*, *16*(3), 267–282. https://doi.org/10.1111/j.1654-1103.2005.tb02365.x
- Walker, D. A., Epstein, H. E., Raynolds, M. K., Kuss, P., Kopecky, M. A., Frost, G. V., et al. (2012). Environment, vegetation and greenness (NDVI) along the North America and Eurasia arctic transects. *Environmental Research Letters*, 7(1), 015504. https://doi.org/10.1088/1748-9326/7/1/015504
- Xu, L., Myneni, R. B., Chapin, F. S. III, Callaghan, T. V., Pinzon, J. E., Tucker, C. J., et al. (2013). Temperature and vegetation seasonality diminishment over northern lands. *Nature Climate Change*, 3(6), 581–586. https://doi.org/10.1038/NCLIMATE1836